Author's response to reviews

Title: The presence of KIR2DS4 full-length gene decreases a chance of rheumatoid arthritis patients to respond to methotrexate treatment

Authors:

Edyta Majorczyk (e.majorczyk@po.opole.pl)
Andrzej Pawlik (pawand@poczta.onet.pl)
Daria Gendosz (dariagendosz@gmial.com)
Piotr Kuśnierczyk (pkusnier@iitd.pan.wroc.pl)

Version: 2
Date: 18 October 2013

Author's response to reviews:

Dear Editors,

my point-by-point response has been provided:

1. Name of ethics committee: Please update your ethics statement to include the name of the ethics committee that approved your study.

Now, the name of the ethics committee (the Committee of Ethical Affairs of the Pomeranian Medical University) has been done.

2. In addition, could you please clarify if the methotrexate treatment was standard patient care and if the researchers had any involvement in the treatment decisions.

It has been changed according to the editor's request.

Sincerely yours,
Edyta Majorczyk